Author's response to reviews

Title: Health systems performance in sub-Saharan Africa: governance, outcome and equity.

Authors:

Anna E Olafsdottir (anna@mikron.is)
Daniel D Reidpath (daniel.reidpath@monash.edu)
Subhash Pokhrel (subhash.pokhrel@brunel.ac.uk)
Pascale Allotey (pascale.allotey@monash.edu)

Version: 4 Date: 11 April 2011

Author's response to reviews:

To BMC Series Editorial Production Team
Please see below the formatting changes made.

Major revisions:
- The author list has been changed according to the preferred style
- Abstract is now the same in the manuscript as in the submission system
- The only footnote in the manuscript was moved to an Endnote section directly before the references. NOTE “footnotes” in the tables were kept there.
- Shading and colours have been removed from tables

Minor revisions:
- Track changes have been removed
- Abstract has been changed as requested
- Vertical lines have been removed from tables
- The manuscript was read through to check for any typographical errors. A few P values were still presented with an accuracy of 3 figures (behind the dot) instead of 2 figures, which was amended according to the reviewers’ requests – must have been a mistake on our behalf. I apology for that (when p=<.001 is though an exception).

With warm regards
Anna Olafsdottir
10th April 2011